Author's response to reviews

Title: Revisiting Interaction in Knowledge Translation

Authors:

Liane R Ginsburg (lgins@yorku.ca)
Steven Lewis (steven.lewis@shaw.ca)
Lisa Zackheim (lisakadish@rogers.com)
Ann Casebeer (alcasebe@ucalgary.ca)

Version: 6 Date: 30 October 2007

Author's response to reviews:

your email of oct 26 asked us to fix:
Affiliations ¿ Please move the country after the postcode in the affiliations. We have done this

References - References should be cited in the text using consecutive numbers in square brackets starting from [1] then [2] etc. References 57, 58, 39 and 56 are cited in incorrect locations on page 9. Please reorder the references so that they are all in consecutive order. WE HAVE DONE THIS. IT REQUIRED REFORMATTING ALL REFERENCES AFTER PAGE 9 BUT IT SHOULD NOW BE CORRECT